Author’s response to reviews

Title: Sexual and reproductive health of women living with HIV in Muslim-majority countries: A systematic mixed studies review

Authors:

Dyah Juliastuti (d.juliastuti@uq.edu.au)

Judith Dean (j.dean4@uq.edu.au)

Lisa Fitzgerald (l.fitzgerald@sph.uq.edu.au)

Version: 2 Date: 09 Sep 2019

Author’s response to reviews:

Dear Editor,

Please accept my revised manuscript. As your suggestion, I sent the manuscript to a professional English language editor in Australia. Please excuse me for not using the suggested editing services. Furthermore, I have done some corrections according to your advise. In regard of your questions, in page 5 line 13, my answer is: Yes the review conducted from that period of time (September 2016 - Feb 2018) in line with my PhD study process.

Please let me know if there is other concern related to the article. Looking forward to hear your feedback. Thank you.

Kind regards,

Dyah